Author's response to reviews

Title: IVC CLAMP: Infrahepatic Inferior Vena Cava Clamping during hepatectomy - A randomised controlled trial in an interdisciplinary setting [NCT00732979]

Authors:

Nuh N. Rahbari (nuh.rahbari@med.uni-heidelberg.de)
Johannes B. Zimmermann (j.zimmermann@med.uni-heidelberg.de)
Moritz Koch (moritz.koch@med.uni-heidelberg.de)
Thomas Bruckner (bruckner@imbi.uni-heidelberg.de)
Thomas Schmidt (thomas.schmidt@chir.ma.uni-heidelberg.de)
Christoph Reissfelder (christoph.reissfelder@med.uni-heidelberg.de)
Markus A. Weigand (markus.Weigand@chiru.med.uni-giessen.de)
Markus W. Büchler (markus.buechler@med.uni-heidelberg.de)
Jürgen Weitz (juergen.weitz@med.uni-heidelberg.de)

Version: 2 Date: 16 January 2009

Author's response to reviews:

Dear Sir or Madam,

we hereby submit the revised version of our manuscript. We thoroughly followed the instructions made by the editorial board.

In detail we added sections on "Competing interests", "Authors' Contributions" and "Acknowledgements".

We critically reviewed the manuscript and made sure that all abbreviations are defined at first use.

We separately sent the ethics documentation to the editorial office.

Please note that there is no external funding for this study.

We are looking forward to your reply. Please do not hesitate to contact us, if there should be any questions.

Kind regards,

Nuh Rahbari